Craig Chambers CSE 401 Introduction to Compiler Construction

Readings and Assignment #4

By: Read:

M 4/17 Reminder: Section 4.9

W 4/19 Section 7.6

F 4/21 Sections 6.1-6.4

M 4/24 Sections 6.5-6.6

By: Do:

M 4/17 Reminder: Exercises 4.1abe, 4.2, 4.3ac, 4.4bcd, 4.11a

W 4/19 Design and implement the necessary extensions to the PL/0 AST class hierarchy to represent the extended PL/0 language. Turn in printouts of your changed implementation, with your changes clearly marked. (I will give you a sample implementation of the AST class hierarchy which you can adopt in place of your version.)

M 4/24 Complete the implementation of the extended PL/0 parser by modifying the parsing routines to parse the extended syntax and construct extended AST trees. As with the scanner project, turn in highlighted printouts of your changed parsing files, test input programs, and output demonstrating that your parser parses the extended language. Run the extended compiler with the “-p -P” options to print the parse tree and then stop.

As usual, you will be graded on correctness of your implementation, on clarity and good design of your implementation, and on sufficiency of your test cases.